Author’s response to reviews

**Title**: Are Proton Pump Inhibitors the First Choice for Acute Treatment of Gastric Ulcers? A Meta Analysis of Randomized Clinical Trials

**Authors:**

Maribel Salas (msalas@caroresearch.com)
Alexandra Ward (alexward@caroresearch.com)
Jaime Caro (icaro@caroresearch.com)

**Version**: 3  **Date**: 10 Jul 2002

Legend text on the figures were deleted. In figure 2 was cropped to minimize the white space.